\documentclass[11pt,reqno]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage[colorlinks]{hyperref}



\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]





%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/#1}{\includegraphics[scale=0.02]{orcid.pdf}}}


\def\udcs{512.542} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{20D30} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}
\newtheorem{pcorollary}{Corollary}

\theoremstyle{definition}
\newtheorem{example}{Example}
\newtheorem{remark}{Remark}



\hypersetup{linkcolor=blue,filecolor=black,urlcolor=blue, citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
     {
  \vbox{
     \noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
     {\LARGE \textsf{СИБИРСКИЕ ЭЛЕКТРОННЫЕ}} \\[2mm]
     {\LARGE \textsf{МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ}} \\[2mm]
     {\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
     {\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
     \raisebox{1ex}{ISSN 1813-3304 }
     \vspace*{-3mm}
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
}
     \vspace{-1mm}
     \noindent
      \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
     \hline\hline
     \end{tabular}

     \vspace{3mm}
     {\flushleft\it Vol. 20, No. 2, pp. 144--145 (2023) \hspace{\fill}{\rm\small УДК \udcs \hspace{-4mm}}}
     \newline  %Volume, pages in Russian, filled by Editorial board
     	{\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2023.20.???}}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaa\;}{\rm\small MSC\ \ \mscs }
  }%\vbox
}




\begin{document}
\begin{thebibliography}{1} % Ссылки на литературу приводятся только на английском языке (даже если у источника нет англоязычной версии).
\bibitem{skiba2023snm}
A.~Liu, M.~Chen, I.N.~Safonova, A.N.~Skiba,
\href{https://doi.org/10.1515/jgth-2023-0064}
{\it Finite groups with modular $\sigma$-subnormal subgroups},
J. Group Theory. 2023.

\bibitem{schmidt1994}
R.~Schmidt
{\it Subgroup Lattices of Groups},
De Gruyter, Berlin, New York, 1994.

\bibitem{skiba2023}
A.M.~Liu, W.~Guo, I.N.~Safonova, A.N.~Skiba,
\href{https://doi.org/10.1007/s00013-023-01884-9}
{Finite groups in which modularity is a transitive relation},
Arch. Math., {\bf 121} (2023), 111--121.

\bibitem{zim89}
I. Zimmermann,
\href{https://doi.org/10.1007/BF01221589}
{Submodular subgroups in finite groups},
Math. Z., {\bf 202} (1989), 545--557.

\bibitem{vasil'ev2015}
V.A.~Vasil’ev,
\href{https://doi.org/10.1134/S0037446615060063}
{Finite groups with submodular Sylow subgroups},
Sib. Math. J. {\bf 56}:6 (2015), 1277--1288.

\bibitem{monakhovsokhor2023}
V.S.~Monakhov, I.L.~Sokhor,
\href{https://doi.org/10.1007/s00013-023-01872-z}
{Finite groups with submodular primary subgroups},
Arch. Math., 121 (2023), 1--10.

\bibitem{kniaginamonakhov2013}
V.N.~Knyagina, V.S.~Monakhov,
\href{https://doi.org/10.1515/dma-2013-011}
{Finite groups with nilpotent and Hall subgroups}
Discrete Math. Appl., {\bf 23}:2 (2013) 175--182.

\bibitem{monakhov2016}
V.S.~Monakhov,
\href{https://doi.org/10.1134/S0037446616020178}
{Finite groups with abnormal and $\mathfrak{U}$-subnormal subgroups},
Sib. Math. J., {\bf 57}:2 (2016), 352--363.

\bibitem{kamornikov2019}
Yi X., Jiang S., Kamornikov S.\,F.
\href{https://doi.org/10.1142/S0219498819500877}
{Finite groups with given non-nilpotent maximal subgroups of prime index},
J. Algebra Appl., {\bf 18}:5 (2019), 1950087-1--1950087-5.

\bibitem{monakhovsokhor2020}
V.S.~Monakhov, I.L.~Sokhor,
\href{https://doi.org/10.22108/IJGT.2018.112602.1500}
{Finite groups with seminormal or abnormal Sylow subgroups},
Int. J. Group Theory, {\bf 9}:3 (2020), 139--142.

\bibitem{ballcl}
A.~Ballester-Bolinches, L.M.~Ezquerro,
{\it Classes of finite groups},
Springer, Dordrecht, 2006.

\bibitem{gap}
\href{https://www.gap-system.org}
{A system for computational discrete algebra GAP 4.12.2}.

\bibitem{vdovin2009}
E.P.~Vdovin,
\href{https://doi.org/10.3103/S1055134409010039}
{Carter subgroups of finite groups},
Sib. Adv. Math., {\bf 19}:1 (2009), 24--74.


\bibitem{huppert1967}
B.~Huppert,
{\it Endliche Gruppen I},
Springer, Berlin, Heidelberg, New York, 1967.

\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}




\thispagestyle{empty}

\title[Finite groups with modular and submodular subgroups]{\large Finite groups with modular and submodular subgroups}


\author[I.L. Sokhor]{{\bf I.L. Sokhor}\protect\orcidicon{0000-0002-7115-4045}}



\address{Irina Leonidovna Sokhor  % Контактные данные всех авторов и место работы указываются только на английском языке
\newline\hphantom{iii} Francisk Skorina Gomel State University,
\newline\hphantom{iii} Kirova Str. 119,
\newline\hphantom{iii} 246019, Gomel, Belarus}%
\email{\textcolor{blue}{irina.sokhor@gmail.com}}%

\thanks{\sc Sokhor, I.L.,
On groups with modular and submodular subgroups}
\thanks{\copyright \ 2023 Sokhor I.L.}
\thanks{\rm The work is supported by  the Ministry of Education of Belarus
(``Convergention-2025'' No. state registr. 20211467)}
\thanks{\it Received January, 1, 2023, Published December, 31, 2023}%


\semrtop \vspace{1cm}
\maketitle {\small

\vspace{-12pt}
\centerline{{\it Communicated by} {\sc P.P. Petrov}
%({\it filled by Editorial board})
}

\bigskip
\bigskip

\begin{quote}
\noindent{\bf Abstract:}
A subgroup $H$ of a group $G$ is modular in $G$ if $H$ is a modular element of
subgroup lattice of $G$, and is submodular in $G$ if there is a subgroup chain
$H=H_0\leq\ldots\leq H_i\leq H_{i+1}\leq \ldots \leq H_n=G$
such that $H_i$ is modular in $H_{i+1}$ for every~$i$.
We prove that if every Sylow subgroup of a group $G$ is modular in $G$,
then $G$ is supersolvable and $G/F(G)$ is a cyclic group of square-free order.
We also obtain new signs of supersolvabilty of groups with some submodular subgroups
(normalizers of Sylow subgroups, Hall subgroups, maximal subgroups).
For a such group $G$, $G/\Phi(G)$ is a supersolvable group of square-free exponent.
Moreover, we describe the structure of groups with modular (submodular) or
self-normalizing primary subgroups.

\medskip

\noindent{\bf Keywords:} finite group, modular subgroup, submodular subgroup, self-normalizing subgroup.
 \end{quote}
}

\bigskip

\section{Introduction}

All groups in this paper are finite.

In many respects, the group structure depends on the way of embedding of its subgroups.
In particular, normal subgroups play an important role in the structure of non-simple groups.
Modularity is an extension of the concept of normality.

\begin{definition}
A subgroup $H$ of a group $G$ is modular in $G$ if $H$ is a modular element of
subgroup lattice of $G$, i.\,e. the following statements hold:

$(1)$~$\langle A, H\rangle \cap B = \langle A, H \cap B \rangle$
for all $A,B \leq G$ such that $A \leq B$;

$(2)$~$\langle A, H\rangle \cap B = \langle H, A \cap B \rangle$
for all $A,B \leq G$ such that $H \leq B$.
\end{definition}

A subgroup $H$ of a group~$G$ is quasinormal in~$G$ if $H$ permutes with every subgroup of~$G$.
Clearly, every quasinormal subgroup, in particular, normal subgroup, is modular.
The converse is not true in general. For example, in the non-abelian group of order $6$,
a subgroup of order $2$ is modular and not quasinormal.
Groups in which every subnormal subgroup is modular have been considered in~\cite{skiba2023snm}.
In~\cite{schmidt1994}, the properties of modular subgroups were described in detail.
Like normality, modularity is not a transitive relation.
In~\cite{skiba2023}, there were obtained a description of finite groups
in which modularity is transitive.

Submodularity is an extension of modularity.

\begin{definition}[{\cite[p.~546]{zim89}}]\label{dfsmod}
A subgroup $H$ of a group $G$ is submodular in $G$ if there is a subgroup chain
\begin{equation}\label{eqkfsn}
H=H_0\leq\ldots\leq H_i\leq H_{i+1}\leq \ldots \leq H_n=G
\end{equation}
such that $H_i$ is modular in $H_{i+1}$ for every~$i$.
\end{definition}

It is clear that every modular subgroup and every subnormal subgroup is submodular.
The converse is not true in general.
Unlike normality and modularity, the relation of submodularity is transitive.
Groups with submodular Sylow subgroup were studied in~\cite{zim89,vasil'ev2015,monakhovsokhor2023}.
We denote the class of all such groups by~$\mathfrak{Z}$.
In~\cite{monakhovsokhor2023}, there were introduced and described the class $\mathfrak{C}$
of all group in which every cyclic subgroup is submodular.
Since a subnormal subgroup is submodular, $\mathfrak{Z}\subset\mathfrak{C}$,
and $C^3_7\rtimes Q\in \mathfrak{C}\setminus \mathfrak{Z}$, where $Q$ is a non-abelian subgroup
of order $3^3$ and exponent~$3$ that acts irreducibly on an elementary abelian group~$C_7 ^3$
of order~$7^3$,~\cite[Example~3]{monakhovsokhor2023}.

In this paper we prove that a group $G$ in which every Sylow subgroup is modular
is supersolvable, and $G/F(G)$ is a cyclic group of square-free order (see Proposition~\ref{pallmod}).
If every Sylow subgroup of a group~$G$ is submodular in $G$, then $G$ can be non-supersolvable 
(see Example~\ref{ex294}). We obtain new signs of supersolvabilty of groups with 
some submodular subgroups (normalizers of Sylow subgroups, Hall subgroups, maximal subgroups).
For a such group $G$, $G/\Phi(G)$ is a supersolvable group of square-free exponent.
Here $F(G)$ and $\Phi(G)$ are the Fitting subgroup and the Frattini subgroup,
respectively.

Sometimes, when we impose restrictions on subgroups,
we have a choice between several possible properties of subgroups.
We say that such restrictions are alternative.
Restrictions can be strictly alternative
when one of the possible subgroup properties excludes all other one,
or partially alternative, otherwise.
For instance, a normal subgroup can not be self-normalizing.
But there are groups that contain self-normalizing modular subgroup, for example,
the non-abelian group of order $6$. Groups with partially alternative restrictions 
on subgroups were studied in papers of many authors,
see \cite{kniaginamonakhov2013,monakhov2016,kamornikov2019,monakhovsokhor2020}
and their references. We describe the structure of groups with modular (submodular) or
self-normalizing primary subgroups (see Theorem~\ref{tmod} and Theorem~\ref{tsmod}).

\section{Preliminaries}

Let $\mathbb{P}$ be the set of all primes. We use $\pi(G)$ to denote the set of all prime divisors of $|G|$.
If $X$ is a subgroup  (proper subgroup, normal subgroup, maximal subgroup) of a group~$Y$,
then we write $X\le Y$ ($X<Y$, $X\lhd Y$, $X\lessdot  \ Y$, respectively).

Let $\mathfrak{F}$ be a formation. The $\mathfrak{F}$-residual of a group~$G$
is the intersection of all normal subgroups of~$G$ with quotients in $\mathfrak{F}$.
It is denoted by~$G^\mathfrak{F}$. We use $\mathfrak{N}$ and $\mathfrak{U}$ to denote
the formation of all nilpotent groups and the formation of all supersolvable groups, respectively.
If $\mathfrak{X}$ is a class of groups, then $\mathfrak{X}_1$ is a class of all groups
from $\mathfrak{X}$ with square-free exponents. In particular, $\mathfrak{U}_1$ is 
the formation of all supersolvable groups of square-free exponent.

Recall a Carter subgroup of a group $G$ is a nilpotent self-normalizing subgroup
of~$G$~\cite[VI.12]{huppert1967}. In solvable groups, Carter subgroups exist and are conjugate.
An insolvable group may have no Carter subgroups, but by a result of E.\,P. Vdovin~\cite{vdovin2009},
which uses the classification of finite simple groups, Carter subgroups are conjugate whenever they exist.

A group $G$ is a $\mathrm{P}$-group~\cite[p.~49]{schmidt1994}
if $G$ is either an elementary abelian group of order $r^{n+1}$ for a prime $r$
or the semidirect product of an elementary abelian normal subgroup $R$ of order $r^n$
and a subgroup~$Q$ of order $q$, $q\neq r$, which induces a nontrivial power automorphism on~$R$.
The properties of such groups are described in~\cite[Lemma~2.2.2]{schmidt1994}.
In particular, every Sylow subgroup of a non-abelian $\mathrm{P}$-group is modular.

The following lemmas contains the properties of modular and submodular subgroups that we use.

\begin{lemma}[{\cite[p.~201]{schmidt1994}, \cite[Lemma~1]{zim89}}]\label{lmsmprop}
Let $G$ be a group, let $H$ be a modular (submodular) subgroup of~$G$,
let $K$ be a subgroup of $G$, and let $N$ be a normal subgroup of $G$.
Then

$(1)$~$H^g$ is modular (submodular)  in $G$ for any $g\in G$;

$(2)$~$H\cap K$ is modular (submodular) in $K$, in particular,
if $H\leq K$, then $H$ is modular (submodular)  in $K$;

$(3)$~ $HN/N$ is modular (submodular) in $G/N$;

$(4)$~if $N\leq K$ and $K/N$ is modular (submodular) in~$G/N$,
then $K$ is modular (submodular) in~$G$.
\end{lemma}

\begin{lemma}\label{lsylmod}
$(1)$~A subgroup $H$ of a group $G$ is quasinormal in $G$ if and only if
$H$ is modular and subnormal in $G$.

$(2)$~Let $Q$ be a Sylow $q$-subgroup of $G$ for a prime $q\in\pi(G)$.
The subgroup $Q$ is modular in $G$ if and only if either $Q$ is normal in $G$
or $Q^G/Q_G$ is a normal Hall subgroup, which is  a non-abelian $P$-group
of order $r^nq$ for primes $r>q$ and has a normal supplement.

$(3)$~Let $A$ be a proper subgroup of a Sylow subgroup of a group $G$.
The subgroup $A$ is modular in $G$ if and only if $A$ is quasinormal in $G$.
In particular, if a primary modular subgroup of $G$ is not a Sylow subgroup of $G$,
then it is subnormal in~$G$.
\end{lemma}

\begin{proof}
$(1)$ This statement is true in view of~\cite[Theorem~5.1.1]{schmidt1994}.

$(2)$~Assume that $Q$ is a Sylow $q$-subgroup of group $G$ for a prime $q\in\pi(G)$
and $Q$ is modular in $G$. If $Q$ is normal in $G$, then $Q$ is modular in $G$.
Suppose that $Q$ is not normal in $G$. In view of Statement~(1),
$Q$ is not quasinormal in $G$. Therefore $G/Q_G=Q^G/Q_G\times K/Q_G$,
where $Q^G/Q_G$ is a non-abelian $P$-group of order $r^nq$ for primes $r>q$,
$Q^G/Q_G$ and $K/Q_G$ of coprimes order, by~\cite[Lemma~5.1.9]{schmidt1994}.
Hence $Q^G/Q_G$ is a normal Hall subgroup and has a normal supplement.

Conversely, if $Q$ is normal in $G$, then $Q$ is modular in $G$ by definition.
Assume that $Q^G/Q_G$ is a normal Hall subgroup, which is a non-abelian $P$-group
of order $r^nq$ for primes $r>q$ and has a normal supplement.
In that case, $Q$ is modular in $G$ by~\cite[Lemma~5.1.9]{schmidt1994}.

$(3)$~Assume that $A$ is a proper subgroup of a Sylow subgroup of~$G$.
If $A$ is modular in $G$, then $A$ is quasinormal in $G$ by~\cite[Lemma~5.1.9]{schmidt1994}.
In particular, $A$ is subnormal in $G$ by Statement~(1).
Conversely, if  $A$ is quasinormal in $G$, then $A$ is modular in $G$ by Statement~(1).
\end{proof}

Notice that a subgroup $M$ of a group $G$ is a maximal modular subgroup of $G$
if $M$ satisfies the following conditions: $M$ is a proper subgroup of $G$;
$M$ is a modular subgroup of $G$; if $M < K < G$, then $K$ is not modular in $G$.

\begin{lemma} \label{l512}
A subgroup $M$ of a group $G$ is a maximal modular subgroup of $G$ if and only if
either $M$ is normal in $G$ and $G/M$ is a simple group
or $M$ is a maximal subgroup of $G$ and $G/M_G$ is a non-abelian group of order $pq$
for primes $p$ and~$q$.
\end{lemma}

\begin{proof}
Let $M$ be a maximal modular subgroup of a group $G$.
Suppose that $M$ is normal in $G$ and $G/M$ is not simple.
In that case, there is a non-trivial normal subgroup $H/M$ of~$G/M$.
Therefore $H$ is normal in~$G$, and $H$ is modular in~$G$.
Since $M< H$, we obtain a contradiction with the choice of~$M$.
Thus, $G/M$ is a simple group. Now assume that $M$ is not normal in $G$.
In that case, $G/M_G$ is a non-abelian group of order $pq$ for primes $p$ and~$q$
by~\cite[Lemma~5.1.2]{schmidt1994}, and $M$ is a maximal subgroup of~$G$.

Conversely, let $M$ be a normal subgroup of $G$ such that $G/M$ is a simple group.
In that case, $M$ is modular in $G$. Suppose that there is a maximal modular subgroup $H$
of $G$ such that $M\leq H$. Since $G/M$ is simple, we deduce that $H$ is not normal in $G$.
Therefore $G/H_G$ is a non-abelian group of order $pq$ for primes $p$ and~$q$
in view of~\cite[Lemma~5.1.2]{schmidt1994}. By the choice of~$M$, we have $M= H_G$
and $G/H_G$ is a simple group, a contradiction.
Now assume that $M$ is a maximal subgroup of~$G$ and $G/M_G$ is a non-abelian group
of order $pq$ for primes $p$ and~$q$. In that case, $M/M_G$ is a maximal subgroup of~$G/M_G$.
Since $|G/M_G|=pq$, we deduce that $M/M_G$ is modular in $G/M_G$.
Hence $M$ is modular in $G$ by Lemma~\ref{lmsmprop}\,(4).
Since $M$ is a maximal subgroup of~$G$, $M$ is a maximal modular subgroup of~$G$.
\end{proof}

\begin{lemma}\label{lrsmod}
Let $R$ be a submodular $r$-subgroup of a group~$G$ for a prime $r\in\pi(G)$.

$(1)$~If $r=\max\pi(G)$, then $R$ is subnormal in $G$ \textup{\cite[Lemma~2.3]{monakhovsokhor2023}}.

$(2)$~If $R$ is a Sylow $r$-subgroup of~$G$, then  $G$ is $r$-solvable
\textup{\cite[Proposition~2.4]{monakhovsokhor2023}}.
\end{lemma}

\begin{lemma}[{\cite[Theorem~1.3]{monakhovsokhor2023}}]\label{lz}
Let~$G$ be a group. The following statements are equivalent.

$(1)$~Every Sylow subgroup of $G$ is submodular in~$G$, i.\,e. $G\in \mathfrak Z$.

$(2)$~$G/\Phi(G)\in \mathfrak Z_1$.

$(3)$~$A/\Phi (A)\in \mathfrak{U}_1$ for any metanilpotent subgroup $A$ of $G$.

$(4)$~$B/\Phi (B)\in \mathfrak{U}_1$ for any biprimary subgroup $B$ of $G$.
\end{lemma}

\begin{lemma}[{\cite[Theorem~1.4]{monakhovsokhor2023}}]\label{lc}
Let~$G$ be a group. The following statements are equivalent.

$(1)$~Every cyclic primary subgroup of~$G$ is submodular in~$G$, i.\,e. $G\in \mathfrak C$.

$(2)$~$G/\Phi(G)\in \mathfrak C_1$.

$(3)$~$A/\Phi (A)\in \mathfrak{U}_1$ for any subgroup $A$ with the nilpotent derived subgroup.

$(4)$~$B/\Phi (B)\in \mathfrak{U}_1$ for any biprimary subgroup $B$ with a cyclic Sylow subgroup.
\end{lemma}


A group $G$ is primitive if $G$ contains a maximal subgroup $M$ such that $M_G=1$.
It is easy to check that the following lemma is true.

\begin{lemma}\label{lprimitivef}
Let $\mathfrak{F}$ be a saturated formation, and let $G$ be a solvable group.
If $G\notin\mathfrak{F}$ and $G/N\in\mathfrak{F}$ for every non-trivial normal subgroup $N$ of~$G$,
then $G$ is a primitive group.
\end{lemma}

The following lemma contains the properties of primitive groups that we use.

\begin{lemma}[{\cite[Theorem~1.1.7,1.1.10]{ballcl}}]\label{lprimitive}
Let $G$ be a solvable primitive group and let $M$ be a maximal subgroup of~$G$ with $M_G=1$.
The following statements hold.

$(1)$~$\Phi(G) = 1$.

$(2)$~$G$ contains a unique minimal normal subgroup~$N$
such that $N = C_G(N)=F(G)=O_p(G)$ for a prime~$p$.

$(3)$~$G=F(G)\rtimes M$ and $O_p(M)=1$.
\end{lemma}



\section{Groups with modular primary subgroups}

\begin{proposition}\label{pallmod}
Let every Sylow subgroup of a group~$G$ be  modular in $G$.
The following statements hold.

$(1)$~$Q^G/Q_G$ is a normal Hall subgroup, which is a $P$-group of order $r^nq$ for primes $r>q$
and has normal supplement for every non-normal Sylow $q$-subgroup $Q$.

$(2)$~$G$ is supersolvable.

$(3)$~$G/F(G)$ is a cyclic group of square-free order.

Conversely, if $G$ satisfies Statement $(1)$,
then every Sylow subgroup of~$G$ is  modular in $G$.
\end{proposition}

\begin{proof}
Assume that every Sylow subgroup of a group~$G$ is  modular in $G$.

$(1)$~If $Q$ is a non-normal Sylow subgroup of~$G$,
then by Lemma~\ref{lsylmod}\,(2), $Q^G/Q_G$ is a normal Hall subgroup,
which is a non-abelian $P$-group of order $r^nq$ for primes $r>q$ and has a normal supplement.

$(2)$~We use induction on $|G|$ to prove that $G$ is supersolvable.
Let $N$ be a non-trivial normal subgroup of~$G$ and let $\overline{P}$ be a Sylow $p$-subgroup
of~$\overline{G}=G/N$ for a prime~$p\in\pi(G/N)$. In that case, there is a Sylow $p$-subgroup $P$
of~$G$ such that $\overline{P}=PN/N$. By hypothesis, $P$ is modular in $G$.
Therefore $\overline{P}$ is modular in $\overline{G}$ by Lemma~\ref{lmsmprop}\,(3).
Consequently, $G/N$ is supersolvable by induction. In view of Lemma~\ref{lprimitivef},
$G$ is primitive, and by Lemma~\ref{lprimitive}, $G=F(G)\rtimes M$, where $F(G)$
is a unique minimal normal subgroup of~$G$, $M$ is a maximal subgroup of $G$ with $M_G=1$.
By Lemma~\ref{lrsmod}\,(1), $F(G)=R$ is a Sylow $r$-subgroup of~$G$ for $r=\max\pi(G)$.
Let $Q$ be a Sylow $q$-subgroup of~$M$. In that case, $Q$ is a Sylow $q$-subgroup of~$G$ and $q< r$.
By hypothesis, $Q$ is modular in $G$. From $Q_G=1$ and Lemma~\ref{lsylmod}\,(2),
it follows that $G=Q^G$ is a non-abelian $P$-group of order $r^nq$ and $M=Q$.
Thus, $Q$ is a non-normal maximal modular subgroup of~$G$. Consequently,
by Lemma~\ref{l512}, we have $|G|=rq$ and  $G$ is supersolvable.

$(3)$~By Statement~(2), $G$ is supersolvable. Therefore $\overline{G}=G/F(G)$ is abelian.
Let $\overline{Q}$ be a Sylow $q$-subgroup of~$\overline{G}$.
In that case, there is a Sylow $q$-subgroup $Q$ of~$G$ such that
$\overline{Q}=QF(G)/F(G)\cong Q/Q\cap F(G)=Q/Q_G$. Since $Q$ is modular in $G$,
by Lemma~\ref{lsylmod}\,(2), $Q^G/Q_G=TQ^G/Q_G\rtimes Q/Q_G$ is a non-abelian $P$-group
of order $t^nq$ for a prime $t>q$. Hence $|\overline{Q}|=|Q/Q_G|=q$, and so
every Sylow subgroup of~$G/F(G)$ is of prime order and $G/F(G)$ is cyclic.

Conversely, let $G$ satisfy Statement $(1)$ and let $Q$ be a Sylow $q$-subgroup of $G$ for a prime $q\in\pi(G)$.
If $Q$ is normal in $G$, then $Q$ is modular in $G$. Suppose that $Q$ is not normal in $G$.
In that case, $Q^G/Q_G$ is a  normal Hall subgroup, which is a non-abelian $P$-group of order $r^nq$
for primes $r>q$ and has a normal supplement. Consequently, $Q$ is modular in $G$ by Lemma~\ref{lsylmod}\,(2).
\end{proof}

\begin{example}
The group $G=C_3\times S_3$~\cite[SmallGroup(18,3)]{gap} is supersolvable,
$F(G)=C_3^2$ and $G/F(G)\cong C_2$. At the same time, a Sylow 2-subgroup $C_2$ is not modular in $G$.
Thus, a group satisfying Statement~$(2)$ and Statement~$(3)$ of Proposition~\ref{pallmod}
may contain a non-modular Sylow subgroup.
\end{example}

\begin{pcorollary}
Every Sylow subgroup of a biprimary group $G$ is modular in $G$ if and only if
either $G$ is nilpotent or $G$ is supersolvable and $G/Q_G$ is a non-abelian $\mathrm{P}$-group
of order $r^nq$, $r>q$, where $Q$ is a Sylow $q$-subgroup of~$G$.
\end{pcorollary}

\begin{proof}
Assume that $G$ is a biprimary group, $Q$ is a Sylow $q$-subgroup of $G$,
$R$ is a Sylow $r$-subgroup of~$G$, $r>q$. By Lemma~\ref{lrsmod}\,(1), $R$ is normal in $G$.
If $Q$ is normal in $G$, then $G$ is nilpotent. Suppose that $Q$ is not normal in $G$.
By Proposition~\ref{pallmod}, $G$ is supersolvable and $G/Q_G$ is a non-abelian $\mathrm{P}$-group
of order $r^nq$.

Conversely, if $G$ is nilpotent, then every Sylow subgroup of~$G$ is normal in $G$.
Hence every Sylow subgroup of~$G$ is modular in $G$. Let $G$ is supersolvable and
$G/Q_G$ is a non-abelian $\mathrm{P}$-group of order $r^nq$, $r>q$.
In that case, every Sylow subgroup of~$G$ is modular in $G$ by Proposition~\ref{pallmod}.
\end{proof}

\begin{remark}
Further, we use $\mathfrak{U}_m$ to denote the class of all supersolvable groups satisfying
Statements~$(1)$--$(3)$ of Proposition~\ref{pallmod}.
\end{remark}

\begin{theorem}\label{tmod}
Every Sylow subgroup of a group $G$ is modular or self-normalizing in $G$ if and only if
either $G\in\mathfrak{U}_m$ or $G=G^\mathfrak{N}\rtimes R$, where
$R$ is a Sylow $r$-subgroup and a Carter subgroup of~$G$ for a prime $r\in \pi(G)$,
$G^\mathfrak{N}\in\mathfrak{N}$.
\end{theorem}

\begin{proof}
Assume that  every Sylow subgroup of a group $G$ is modular or self-normalizing in $G$.
If every Sylow subgroup of $G$ is modular in $G$, then $G\in\mathfrak{U}_m$ by Proposition~\ref{pallmod}.
Now, we can assume that $G\notin\mathfrak{U}_m$. In that case, $G$ contains a Sylow $r$-subgroup $R$
that is not modular in $G$. By hypothesis, $R=N_G(R)$, and $R$ is a Carter subgroup of~$G$.
Let $Q$ be a Sylow $q$-subgroup of~$G$ for a prime $q\neq r$.
If $Q$ is self-normalizing in $G$, then $Q$ is a Carter subgroup of~$G$.
By~\cite{vdovin2009}, Carter subgroups are conjugate whenever they exist.
Hence $R$ and $Q$ are conjugate, a contradiction.
Therefore all Sylow $r'$-subgroups of $G$ are modular in $G$.
If $Q$ is not normal in $G$, then by Lemma~\ref{lsylmod}\,(2),
$G/Q_G=Q^G/Q_G\times K/Q_G$ and  $(|Q^G/Q_G|,|K/Q_G|)=1$.
Since $r\neq q$, we have $RQ_G/Q_G\leq Q^G/Q_G$ or $RQ_G/Q_G\leq K/Q_G$.
In view of~\cite[Theorem~9.3]{vdovin2009}, $RQ_G/Q_G$ is a Carter subgroup of~$G/Q_G$.
Hence $RQ_G/Q_G=Q^G/Q_G$ or $K/Q_G=G/Q_G$, a contradiction, since $Q\neq Q_G$.
Thus, all Sylow $r'$-subgroups of $G$ are  normal in $G$. Therefore a Hall $r'$-subgroup $G_{r'}$
of $G$ is nilpotent  and normal in $G$, $G$ is solvable and $G=G_{r'}\rtimes R$.
Since $G/G_{r'}\cong R\in\mathfrak{N}$, we get $G^\mathfrak{N}\leq G_{r'}$.
Suppose that $G^\mathfrak{N}\rtimes R\leq M\lessdot G$. In that case,
$M/G^\mathfrak{N}$ is a maximal subgroup of the nilpotent group $G/G^\mathfrak{N}$.
Consequently, $M/G^\mathfrak{N}$ is normal in $G/G^\mathfrak{N}$,
and $M$ is normal in $G$. Since $R=N_G(R)$, we get $M=N_G(M)$, a contradiction.
Hence $G=G^\mathfrak{N}\rtimes R$ and $G^\mathfrak{N}=G_{r'}\in\mathfrak{N}$.

Conversely, if $G\in\mathfrak{U}_m$, then every Sylow subgroup of~$G$ is modular in~$G$
by Proposition~\ref{pallmod}. Let $G=G^\mathfrak{N}\rtimes R$, where $R$ is a Sylow $r$-subgroup
and a Carter subgroup of~$G$ for a prime $r\in \pi(G)$, $G^\mathfrak{N}\in\mathfrak{N}$.
Choose an arbitrary Sylow $q$-subgroup $Q$ of~$G$. If $q=r$, then $Q=R^g$ for some element $q\in G$
and $Q$ is self-normalizing in $G$. Let $q\neq r$. In that case, $Q\leq G^\mathfrak{N}\in\mathfrak{N}$,
and $Q$ is normal in $G^\mathfrak{N}$. Consequently, $Q$ is normal in $G$, and $Q$ is modular in $G$.
\end{proof}

A subgroup~$H$ of a group~$G$ is abnormal in $G$ if $g\in\langle H, H^g\rangle$ for every $g\in G$.
Since a primary subgroup is abnormal if and only if it is self-normalizing,
we obtain the following corollary.

\begin{corollary}
If every Sylow subgroup of a group $G$ is modular or self-normalizing in $G$,
then every non-abnormal Sylow subgroup of~$G$ is normal in $G$.
\end{corollary}

\begin{corollary}
Let $G$ be a biprimary group, let $Q$ be a Sylow $q$-subgroup of $G$,
and let $R$  be a Sylow $r$-subgroup of $G$, $r>q$.
Every Sylow subgroup of $G$ is modular or self-normalizing in $G$
if and only if $G$ is a group of one of the following types:

$(1)$ $G$ is nilpotent;

$(2)$ $G$ is supersolvable and $G/Q_G$ is a non-abelian $\mathrm{P}$-group of order $r^nq$;

$(3)$ $G=R\rtimes Q$, $G^\mathfrak{N}=R$ and $Q$ is a Carter subgroup of $G$.
\end{corollary}

\begin{example}
In $G=C_3^3\rtimes A_4$~\cite[SmallGroup(324,160)]{gap},
a Sylow $3$-subgroup $C_3\wr C_3$ is maximal and self-normalizing in $G$,
a Sylow $2$-subgroup $C_2^2$ is contained in a normal maximal subgroup $C_3^3\rtimes C_2^2$.
Since $C_3^3\rtimes C_2^2\in\mathfrak{U}_1$, we conclude that $C_2^2$ is submodular in $C_3^3\rtimes C_2^2$.
Consequently, $C_2^2$ is submodular in~$G$. At the same time,
$C_2^2$ is not normal in $G$. Therefore  in Theorem~\ref{tmod}, the condition of modularity
of Sylow subgroups can not be weakened to submodularity.
\end{example}


\begin{corollary}\label{cmod}
Every primary subgroup of a group $G$ is modular or self-normalizing in $G$
if and only if  $G$ is a  group of one of the following types:

$(1)$~$G\in\mathfrak{U}_m$ and every primary subgroup, which is not a Sylow subgroup of $G$,
is quasinormal in $G$;

$(2)$~$G=G^\mathfrak{N}\rtimes R$, where $R$ is a Sylow $r$-subgroup and a Carter subgroup of $G$
for a prime $r\in \pi(G)$, $G^\mathfrak{N}\times X\in\mathfrak{N}$  for any proper subgroup $X$ of $R$
and all primary subgroups except for Carter subgroups are quasinormal in $G$.
\end{corollary}

\begin{proof}
Assume that every primary subgroup of $G$ is modular or self-normalizing in $G$.
By Theorem~\ref{tmod}, either $G\in\mathfrak{U}_m$ or  $G=G^\mathfrak{N}\rtimes R$, where
$R$ is a Sylow $r$-subgroup and a Carter subgroup of~$G$ for a prime $r\in \pi(G)$,
$G^\mathfrak{N}\in\mathfrak{N}$. Let $G\in \mathfrak{U}_m$ and let $X$ be a primary subgroup of~$G$,
which is not a Sylow subgroup of $G$. In that case, $X$ is modular in $G$,
and in view of Lemma~\ref{lsylmod}\,(3), $X$ is quasinormal in $G$.
Now assume that $G=G^\mathfrak{N}\rtimes R$, where
$R$ is a Sylow $r$-subgroup and a Carter subgroup of~$G$ for a p rime $r\in \pi(G)$,
$G^\mathfrak{N}\in\mathfrak{N}$. Let $X$  be a proper subgroup of~$R$.
If $X$ is self-normalizing in $G$, then $X$ is a Carter subgroup of~$G$.
Hence $X$ and $R$ are conjugate, a contradiction. Therefore $X$ is modular in $G$.
Since $X<R$, we deduce that $X$ is quasinormal in $G$ by Lemma~\ref{lsylmod}\,(3),
in particular, $X$ is subnormal in $G$. Hence $X$ is normal in $G^\mathfrak{N}$
and $G^\mathfrak{N}X=G^\mathfrak{N}\rtimes X\in\mathfrak{N}$.
Let $A$ be a $q$-subgroup of $G$ for a prime $q\neq r$.
If $A$ is a Sylow subgroup of~$G$, then  $A\leq G^\mathfrak{N}\in\mathfrak{N}$.
Consequently, $A$ is normal in $G^\mathfrak{N}$, and $A$ is normal in $G$,
in particular, $A$ is quasinormal in $G$. Suppose that $A$ is not a Sylow subgroup of~$G$.
In that case, $A$ is modular in $G$. Indeed, if $A$ is not modular in $G$, then
$A$ is  self-normalizing in $G$ by hypothesis, and $A$ is a Carter subgroup.
Consequently, $A$ and $R$ are conjugate, a contradiction.
Therefore $A$ is quasinormal in $G$ by Lemma~\ref{lsylmod}\,(3).

Conversely, let $G\in\mathfrak{U}_m$ and let every primary subgroup, which is not a Sylow subgroup of $G$,
be quasinormal in $G$. In that case, in view of Proposition~\ref{pallmod},
every Sylow subgroup of~$G$ is modular in $G$, and by Lemma~\ref{lsylmod}\,(1), every primary subgroup,
which is not a Sylow subgroup of $G$, is modular in $G$. Thus, every primary subgroup of~$G$
is modular in~$G$. Now, assume that  $G=G^\mathfrak{N}\rtimes R$, where
$R$ is a Sylow $r$-subgroup and a Carter subgroup of~$G$ for a prime $r\in \pi(G)$,
$G^\mathfrak{N}\rtimes X\in\mathfrak{N}$ for any proper subgroup $X$ of $R$ and
every primary subgroup except for Carter subgroups is quasinormal in $G$.
By Lemma~\ref{lsylmod}\,(1),  every primary subgroup except for Carter subgroups is modular in $G$.
All Carter subgroups are self-normalizing in $G$.
\end{proof}

\begin{remark}
In view of~\cite[Lemma~2.3.2]{schmidt1994}, a primary group has modular subgroup lattice
if and only if any two of its subgroups permute. Therefore if a group $G$ satisfies Statement $(2)$
of Corollary~\ref{cmod} and $A$ is a $q$-subgroup of $G$ that is not a Carter subgroup, then $A$
has modular subgroup lattice. By~\cite[Theorem~2.3.1]{schmidt1994},
$A$ is a  group of one of the following types:

$(a)$~$A=Q_8\times C_2^n$ for some $n$;
%G is a direct product of a quaternion group Q8 of order 8 with an elementary abelian 2-group, or

$(b)$~$A$ contains an abelian normal subgroup $N$ such that $A/N$ is cyclic;
further there exists an element $x\in A$, $A=N\langle x\rangle$, and  a positive integer $s$
such that $x^{-1}yx=y^{1-q^s}$ for all $y\in N$, with $s\geq 2$ in case $q=2$.
\end{remark}

\begin{corollary}
Assume that every primary subgroup of a group $G$ is  modular or self-normalizing in $G$,
$K$ is a Carter subgroup of~$G$ and $A$ is a primary subgroup of~$G$.
If $|K|$ divides $|A|$, then $A$ is  abnormal in $G$.
If $|K|$ does not divide $|A|$, then $A$ is quasinormal in $G$.
\end{corollary}



\section{Groups with submodular subgroups}

\begin{example}\label{ex294}
In $G=C_7^2\rtimes S_3$ \cite[{SmallGroup(294,7)}]{gap},
a Sylow $7$-subgroup $R=C_7^2$ is normal. Hence  $R$ is modular in $G$.

For a Sylow $3$-subgroup $Q=C_3$, there is a subgroup chain
$$
Q=C_3\lessdot Q_1=C_7\rtimes C_3\lessdot Q_2=C_7^2\rtimes C_3\lessdot G.
$$
Since $Q_1=C_7\rtimes C_3$, we deduce that every subgroup of $Q_1$ is modular in $Q_1$,
in particular, $Q$ is modular in $Q_1$. We have ${(Q_1)}_{Q_2}\cong C_7$ and
$Q_2/{(Q_1)}_{Q_2}\cong C_7\rtimes C_3$. Therefore $Q_1$ is modular in $Q_2$.
Since $Q_2$ is normal in $G$, we get $Q_2$ is modular in $G$.
Thus, $Q$ is submodular in~$G$.

For a Sylow $2$-subgroup $P=C_2$, there is a subgroup chain
$$
P=C_2 \lessdot P_1=C_{14}\lessdot P_2=C_7\rtimes D_{14}\lessdot G.
$$
Since $P$ is normal in $P_1$, we get $P$ is modular in $P_1$.
We have ${(P_1)}_{P_2}\cong C_7$ and $P_2/{(P_1)}_{P_2}\cong D_{14}$.
Hence $P_1$ is modularа in $P_2$. From ${(P_2)}_G\cong C_7^2$ and $G/{(P_2)}_G\cong S_3$,
it follows that $P_2$ is modular in $G$.
Thus,  $P$ is submodular in~$G$ and $G\in\mathfrak{Z}$.
But $G$ is not supersolvable. Therefore $\mathfrak{Z}\not \subseteq \mathfrak{U}$.
\end{example}

\begin{proposition}\label{psmod}
Let $G$ be a group. Then $G/\Phi(G)\in\mathfrak{U}_1$ in every of the following cases.

$(1)$~The normalizer of every Sylow subgroup of $G$ is submodular in $G$.

$(2)$~Every Hall subgroup of $G$ is submodular in $G$.

$(3)$~Every maximal subgroup of $G$ is submodular in $G$.
\end{proposition}

\begin{proof}
$(1)$~Since every subgroup is normal in own normalizer, every Sylow subgroup of $G$
is submodular in $G$. Consequently, $G\in \mathfrak{Z}$, in particular,
$G$ is solvable by Lemma~\ref{lrsmod}\,(2) and $G/\Phi(G)\in \mathfrak Z_1$ by Lemma~\ref{lz}.

Now we prove that $G\in\mathfrak{U}$. Let $N$ be a non-trivial normal subgroup of~$G$
and let $\overline{P}$ be a Sylow $p$-subgroup of~$\overline{G}=G/N$ for a prime $p\in\pi(G/N)$.
In that case, there is a Sylow $p$-subgroup $P$ of~$G$  such that $\overline{P}=PN/N$,
and $N_{\overline{G}}(\overline{P})=N_{G/N}(PN/N)=N_G(P)N/N$. Since $N_G(P)$ is submodular in $G$,
we deduce that $N_G(P)N/N$ is submodular in $G/N$ by Lemma~\ref{lmsmprop}\,(3).
Consequently, by induction, $G/N\in\mathfrak{U}$ for every non-trivial normal subgroup~$N$.
Hence $G$ is primitive by Lemma~\ref{lprimitivef}, and in view of Lemma~\ref{lprimitive},
$G=F(G)\rtimes M$, where $M$ is a maximal subgroup of~$G$ with $M_G=1$.
In view of Lemma~\ref{lrsmod}\,(2), a Sylow $r$-subgroup $R$ of $G$ is
normal in $G$ for $r=\max \pi(G)$. Therefore $R=F(G)$ and $G=R\rtimes M$.
By induction, $G/R\cong M\in \mathfrak{U}$. Consequently, for $q=\max \pi(M)$,
a Sylow $q$-subgroup $Q$ of $M$ is normal in $M$, and so $N_M(Q)=M\leq N_G(Q)\leq G$ and $M=N_G(Q)$.
Since $M$ is a Hall subgroup of $G$, we  deduce that $Q$ is a Sylow subgroup of~$G$.
Hence $M$ is submodular in $G$, and $|R|=|G:M|\in \mathbb{P}$ in view of Lemma~\ref{l512}.
Consequently, $G$ is supersolvable, a contradiction.

Thus, $G/\Phi(G)\in\mathfrak{Z}_1\cap\mathfrak{U}\subseteq \mathfrak{U}_1$.

$(2)$~It is clear that $G\in \mathfrak{Z}$, in particular, $G$ is solvable by Lemma~\ref{lrsmod}\,(2)
and $G/\Phi(G)\in \mathfrak Z_1$ by Lemma~\ref{lz}.

Now we prove that $G\in\mathfrak{U}$. Let $N$ be a non-trivial normal subgroup of~$G$
and let $\overline{H}$ be a Hall subgroup of~$\overline{G}=G/N$.
In that case, there is a Hall subgroup $H$ of $G$ such that $\overline{H}=HN/N$.
Since $H$ is submodular in $G$, we get $HN/N$ is submodular in $G/N$ by Lemma~\ref{lmsmprop}\,(3).
By induction, $G/N\in\mathfrak{U}$ for every non-trivial normal subgroup $N$.
Therefore $G$ is primitive by Lemma~\ref{lprimitivef},
and $G=F(G)\rtimes M$, where $M$ is a maximal subgroup of~$G$ with $M_G=1$, by Lemma~\ref{lprimitive}.
In view of Lemma~\ref{lrsmod}\,(1), a Sylow $r$-subgroup $R$ is normal in $G$ for $r=\max \pi(G)$.
Hence $R=F(G)$ and $G=R\rtimes M$. By induction, $G/R\cong M\in \mathfrak{U}$.
Since $M$ is a Hall subgroup of $G$, we have that $M$ is submodular in $G$.
It follows that $|R|=|G:M|\in \mathbb{P}$ by Lemma~\ref{l512},
and $G$ is supersolvable, a contradiction.

Thus, $G/\Phi(G)\in\mathfrak{Z}_1\cap\mathfrak{U}\subseteq \mathfrak{U}_1$.

$(3)$~Let $M$ be a maximal subgroup of $G$. By hypothesis,  $M$ is modular in $G$.
If $M$ is normal in $G$, then $M=M_G$ and $|G/M_G|\in\mathbb{P}$. Hence  $G/M_G\in \mathfrak{U}_1$.
If $M$ is not normal in $G$, then   $G/M_G\in \mathfrak{U}_1$ by Lemma~\ref{l512}.
Thus, $G/M_G\in \mathfrak{U}_1$ for every maximal subgroup $M$ of~$G$.
Therefore $G/\Phi(G)=G/\bigcap_{M\lessdot G}M_G=\bigcap_{M\lessdot G}G/M_G\in\mathfrak{U}_1$.
\end{proof}

\begin{example}
The Frobenius group $F_7=C_7\rtimes C_6\in\mathfrak{U}_1$, $\Phi(G)=1$, but a maximal subgroup $C_6$
is not submodular in $F_7$. Therefore the statements of Proposition~\ref{psmod} are not converse.
\end{example}

\begin{example}
In the cyclic group $C_{12}$ of order 12, every subgroup is normal, and so is submodular,
but $\exp (C_{12})=12$ and $G\notin\mathfrak{U}_1$. Therefore  
we can not replace~``$G/\Phi(G)\in\mathfrak{U}_1$'' by~``$G\in\mathfrak{U}_1$''
in Proposition~\ref{psmod}.
\end{example}

\begin{theorem}\label{tsmod}
Every primary subgroup of a group $G$ is submodular or self-normalizing in $G$
if and only if $G$ is a  group of one of the following types:

$(1)$~$G\in\mathfrak{Z}$;


$(2)$~$G\notin\mathfrak{C}$ and $G=G'\rtimes\langle x\rangle$,
where $\langle x\rangle$ is a Sylow $r$-subgroup and a Carter subgroup of~$G$,
$r\in\pi(G)$, $G'\rtimes\langle x^r\rangle\in\mathfrak{Z}$;

$(3)$~$G\in\mathfrak{C}\setminus\mathfrak{Z}$ and $G=G^\mathfrak{N}\rtimes R$,
where $R$ is a non-submodular non-cyclic Sylow $r$-subgroup and a Carter subgroup of~$G$,
$r=\min \pi(G)$, $G^\mathfrak{N}\rtimes X\in\mathfrak{Z}$ for every $X < R$.
\end{theorem}

\begin{proof}
Let every primary subgroup of a group $G$ be submodular or self-normalizing in $G$.
If every primary subgroup of~$G$ is submodular in $G$, then $G\in\mathfrak{Z}$.
Further, we can assume that  $G\notin\mathfrak{Z}$. In that case, $G$ contains
a primary subgroup $R$ that is not submodular in $G$. By hypothesis, $R=N_G(R)$, 
and so $R$ is a Sylow $r$-subgroup and a Carter subgroup of $G$ for a prime $r\in\pi(G)$. 
Let $Q$ be a Sylow $q$-subgroup of~$G$ for a prime $q\neq r$. 
If $Q$ is self-normalizing in $G$, then $Q$ a Carter subgroup of~$G$.
Since Carter subgroups are conjugate whenever they exist by~\cite{vdovin2009},
$R$ and $Q$ are conjugate, a contradiction. Therefore all Sylow $r'$-subgroups
of $G$ are modular in $G$. By Lemma~\ref{lrsmod}\,(2), $G$ is $q$-solvable for 
every $q\in\pi=\pi(G)\setminus\{r\}$. Hence $G$ is $\pi$-solvable. 
Since $|\pi(G)\setminus \pi|=|\{r\}|=1$, $G$ is solvable.
Consequently, $G=G^\mathfrak{N}R$ and $G_{r'}\leq G^\mathfrak{N}$.

Let $X$ be a proper subgroup of $R$ and let $Y$ be a primary subgroup of~$G^\mathfrak{N}X$.
If $Y$ is self-normalizing in $G$, then $Y$ is a Carter subgroup of~$G$,
and so $Y$ and $R$ are conjugate, a contradiction. Consequently, every primary subgroup
of~$G^\mathfrak{N} X$ is submodular in $G^\mathfrak{N} X$ by Lemma~\ref{lmsmprop}\,(2)
and $G^\mathfrak{N} X\in\mathfrak{Z}$.

Assume that $G\in\mathfrak{C}$. In that case, $R$ is not cyclic.
By Lemma~\ref{lrsmod}, $G$ has a Sylow tower of supersolvable type.
Hence $r=\min\pi(G)$ and there is a normal Hall $r'$-subgroup $G_{r'}$ of $G$.
Since $G/G_{r'}\cong R\in\mathfrak{N}$, we have $G^\mathfrak{N}\leq G_{r'}$,
and so $G^\mathfrak{N}=G_{r'}$ and $G=G^\mathfrak{N}\rtimes R$.

Now assume that $G\notin\mathfrak{C}$. In that case, $R=\langle x\rangle$  is cyclic.
By~\cite[IV.2.6]{huppert1967}, there is a normal Hall $r'$-subgroup $G_{r'}$ of $G$.
Since $G/G_{r'}\cong R\in\mathfrak{A}$, we have $G^\mathfrak{N}\leq G'\leq G_{r'}$.
Hence $G'=G^\mathfrak{N}=G_{r'}$ and $G=G'\rtimes \langle x\rangle$.
Since $G^\mathfrak{N}\rtimes X\in\mathfrak{Z}$ for every $X<R$,
we deduce that $G'\rtimes \langle x^r\rangle\in\mathfrak{Z}$.

Conversely, if $G\in\mathfrak{Z}$, then every primary subgroup of~$G$ is submodular in~$G$.
Now, we can assume that $G\notin\mathfrak{Z}$. Let $G\notin\mathfrak{C}$ and let $G=G'\rtimes\langle x\rangle$,
where $\langle x\rangle$ is a Sylow $r$-subgroup and a Carter subgroup of $G$,
$r\in\pi(G)$, $G'\rtimes\langle x^r\rangle\in\mathfrak{Z}$.
Choose an arbitrary $q$-subgroup $A$ of $G$ for a prime $q\in\pi(G)$.
If $q\neq r$, then $A\leq G'$. From $G'\leq G'\rtimes\langle x^r\rangle\in \mathfrak{Z}$,
it follows that $A$ is submodular in $G'$.  Since $G'$ is submodular in $G$,
$Q$ is submodular in $G$. Let $q=r$. In that case, $Q^g\leq\langle x\rangle$
for $g\in G$. If $Q^g=\langle x\rangle$, then $Q$ is self-normalizing.
If $Q^g<\langle x\rangle$, then $Q^g\leq\langle x^r\rangle\leq G'\rtimes\langle x^r\rangle\in \mathfrak{Z}$.
Consequently, $Q^g$ is submodular in $G'\rtimes\langle x^r\rangle$,
and $G'\rtimes\langle x^r\rangle$ is submodular in $G$.
Thus, $Q$ is submodular in $G$ by Lemma~\ref{lmsmprop}\,(1).
Now, let $G\in\mathfrak{C}\setminus\mathfrak{Z}$ and let $G=G^\mathfrak{N}\rtimes R$,
where $R$ is a non-submodular non-cyclic Sylow $r$-subgroup and a Carter subgroup of $G$,
$r=\min \pi(G)$, $G^\mathfrak{N}\rtimes X\in\mathfrak{Z}$ for every $X < R$.
As above we can prove that every primary subgroup of~$G$ is submodular or self-normalizing in~$G$.
\end{proof}



\bigskip

\end{document}
